Author's response to reviews

Title: Emergency vaccination of rabies under limited resources - combating or containing?

Authors:

Dirk Eisinger (dirk.eisinger@ufz.de)
Hans-Hermann Thulke (hans.thulke@ufz.de)
Thomas Selhorst (thomas.selhorst@wus.bfav.de)
Thomas Mueller (thomas.mueller@wus.bfav.de)

Version: 4 Date: 8 February 2005

Author's response to reviews: see over
Changes within Resubmission 4 of:

Article title: Emergency vaccination of rabies under limited resources - combating or containing?
MS ID : 1161983425409182
Authors : Dirk Eisinger, Hans-Hermann Thulke, Thomas Selhorst and Thomas Mueller
Journal : BMC Infectious Diseases

We adjusted two spelling errors:
- Now using “threat” instead of “thread” twice on page 11.

Additionally
- We changed “in the US” into “in North-America” in both background sections, because we were reminded, that major advances against rabies in wild-life were achieved in Canada last years.

Finally
- We removed the redundant headline “References” from the text.